Author's response to reviews

Title: The Mental Vitality at Work Study: Design of a Randomized Controlled Trial on the Effect of a Workers' Health Surveillance Mental Module for Nurses and Allied Health Professionals

Authors:

- Fania R Gärtner (F.R.Gartner@amc.uva.nl)
- Sarah M Ketelaar (S.M.Ketelaar@amc.uva.nl)
- Odile Smeets (OSmeets@trimbos.nl)
- Linda Bolier (LBolier@trimbos.nl)
- Eva Fischer (EFischer@trimbos.nl)
- Frank JH van Dijk (F.J.vanDijk@amc.uva.nl)
- Karen Nieuwenhuijsen (K.Nieuwenhuijsen@amc.uva.nl)
- Judith K Sluiter (J.Sluiter@amc.uva.nl)

Version: 2 Date: 3 May 2011

Author's response to reviews:

Dear Editor,

Thank you very much for reviewing our manuscript entitled “The Mental Vitality @ Work Study: Design of a Randomized Controlled Trial on the Effect of a Workers’ Health Surveillance Mental Module for Nurses and Allied Health Professionals”

(MS: 1338344316540811). We are very pleased to hear that our paper is accepted for publication.

We have adapted the manuscript in line with all format changes that were required and did a final proof reading. The adapted version of our manuscript is uploaded in the submission system.

Thank you very much again for the opportunity to publish our paper to in BMC Public Health.

Yours sincerely, on behalf of all authors,

Fania Gärtner